Author’s response to reviews

Title: Mentorship needs at academic institutions in resource-limited settings: A survey at Makerere University College of health sciences

Authors:

- Damalie Nakanjako (drdamalie@yahoo.com)
- Pauline Byakika-Kibwika (pbyakika@gmail.com)
- Kenneth Kintu (kkintu@mujhu.org)
- Jim Aizire (jaizire@mujhu.org)
- Fred Nakwagala (nakwagala@yahoo.com)
- Simon Luzige (drsimluz@hotmail.com)
- Charles Namisi (namisipc@yahoo.co.uk)
- Harriet Mayanja-Kizza (hmk@chs.mak.ac.ug)
- Moses R. (mkamya@infocom.co.ug)

Version: 4 Date: 9 June 2011

Author’s response to reviews:

Response to reviewer’s comments

Age and gender of mentors has been included in Table 1 and in the results section on page 7 under the subheading

Age and gender of mentees were included in the narrative results on mentees on Page 7.

The changes are in bold

Authors have edited the manuscript for English and typos.